Author's response to reviews

Title: A 'snip' in time: What is the best age to circumcise?

Authors:

Brian J Morris (brian.morris@sydney.edu.au)
Jake H Waskett (jake@waskett.org)
Joya Banerjee (joya.banerjee@gmail.com)
Richard G Wamai (r.wamai@neu.edu)
Aaaron AR Tobian (atobian1@jhmi.edu)
Ronald H Gray (RGray@jhsp.h.edu)
Stefan A Bailis (sabailis@mindspring.com)
Robert C Bailey (rcbailey@uic.edu)
Jeffrey D Klausner (drklausner@hotmail.com)
Robin J Willcourt (rjwill246@me.com)
Daniel T Halperin (dhalpe@gmail.com)
Thomas E Wiswell (thomas_wiswell@yahoo.com)
Adrian Mindel (adrian.mindel@sydney.edu.au)

Version: 3 Date: 12 December 2011

Author's response to reviews:

Dr Halperin advises a change of address to:

One of the authors wanted very minor modification to Response to reviewer's comment 5a. Our Response has therefore changed to:

OUR RESPONSE: We agree, and have now added on p. 12, in the 1st paragraph of Conclusions the following:

Para 1, lines 2-3: “when performed by an experienced operator in a clinical or other appropriate setting”.

Para 3, lines 5-8: “The procedure should be performed by a trained professional using appropriate local anesthesia in a clean environment. Circumcision outside of such a setting is ill-advised, so explaining why clinical MC is increasingly being made available in European countries to Muslim families.”

A new ref has been added on p. 9, para 4, line 20:

155a

The new reference in References section is: